Author's response to reviews

Title: The validation of a new measure quantifying the social quality of life of ethnically diverse older women: two cross-sectional studies

Authors:

Luciana Lagana` (LLASPPHI@aol.com)
Maria L. Bratly (mariabratly@ymail.com)
Ioakim Boutakidis (iboutakidis@exchange.fullerton.edu)

Version: 8 Date: 12 September 2011

Author's response to reviews:

Dear Dr. Neilan,

We have made several small yet needed changes to the manuscript (e.g., deleting unnecessary words, repositioning portions of some sentences, abbreviating journals’ titles, and editing punctuation); please post this final version on your Journal’s website. Among other edits, the new title is “The validation of a new measure quantifying the social quality of life of ethnically diverse older women: two cross-sectional studies.” Also, we have italicized the names of the main variables in all the tables.

Thank you very much for your support and patience.

Warm regards,

Luciana Lagana`